

# **JGR** Oceans

## **RESEARCH ARTICLE**

10.1029/2021JC017427

#### **Key Points:**

- An increasing trend in marine heatwaves is observed over the western Indian Ocean and the Bay of Bengal
- Ocean warming and El Niño events drives the genesis and trends of marine heatwaves in the Indian Ocean
- Marine heatwaves reduce the monsoon rainfall over the central Indian subcontinent while enhancing it over the southern peninsula

**Supporting Information:** 

Supporting Information may be found in the online version of this article.

**Correspondence to:** 

M. K. Roxy, roxy@tropmet.res.in

#### Citation:

Saranya, J. S., Roxy, M. K., Dasgupta, P., & Anand, A. (2022). Genesis and trends in marine heatwaves over the tropical Indian Ocean and their interaction with the Indian summer monsoon. *Journal* of Geophysical Research: Oceans, 127, e2021JC017427. https://doi. org/10.1029/2021JC017427

Received 1 APR 2021 Accepted 11 JAN 2022

#### **Author Contributions:**

Conceptualization: M. K. Roxy Data curation: J. S. Saranya Formal analysis: J. S. Saranya Visualization: J. S. Saranya, M. K. Roxy Writing – original draft: J. S. Saranya, M. K. Roxy Writing – review & editing: Panini Dasgupta, Ajay Anand

© 2022. American Geophysical Union. All Rights Reserved.

# Genesis and Trends in Marine Heatwaves Over the Tropical Indian Ocean and Their Interaction With the Indian Summer Monsoon

## J. S. Saranya<sup>1,2</sup>, M. K. Roxy<sup>1</sup>, Panini Dasgupta<sup>1,3</sup>, and Ajay Anand<sup>1,4</sup>

<sup>1</sup>Centre for Climate Change Research, Indian Institute of Tropical Meteorology, Pune, India, <sup>2</sup>College of Climate Change and Environmental Sciences, Kerala Agricultural University, Thrissur, India, <sup>3</sup>Department of Meteorology and Oceanography, College of Science and Technology, Andhra University, Visakhapatnam, India, <sup>4</sup>Department of Atmospheric Sciences, Cochin University of Science and Technology, Cochin, India

**Abstract** Marine heatwaves (MHWs) are extreme oceanic warm water events (above 90th percentile threshold) that significantly impact the marine environment. Recent studies have explored the genesis and impacts of MHWs in the global oceans though they are least understood in the tropical Indian Ocean. Here we investigate the genesis and trend of MHWs in the Indian Ocean during 1982–2018 and their role in modulating the Indian summer monsoon. We find that the rapid warming in the Indian Ocean plays a critical role in increasing the number of MHWs. Meanwhile, the El Niño has a prominent influence on the occurrence of MHWs during the monsoon. The Indian Ocean warming and the El Niño variability have synergistically resulted in some of the strongest and long-lasting MHWs in the Indian Ocean. The western Indian Ocean region experienced the largest increase in MHWs at a rate of 1.2–1.5 events per decade, followed by the north Bay of Bengal at a rate of 0.4–0.5 events per decade. Locally, the MHWs are induced by increased solar radiation, relaxation of winds, and reduced evaporative cooling. In the western Indian Ocean, the decreased winds further restrict the heat transport by ocean currents from the near-equatorial regions toward the north. Our analysis indicates that the MHWs in the western Indian Ocean and the north Bay of Bengal lead to a reduction in monsoon rainfall over the central Indian subcontinent. On the other hand, there is an enhancement of monsoon rainfall over southwest India due to the MHWs in the Bay of Bengal.

**Plain Language Summary** Marine heatwaves are periods of extremely high temperatures in the ocean. Though recent studies have reported their occurrence and impacts in the global oceans, they are least understood in the tropical Indian Ocean. In the current study, we find that marine heatwaves are increasing in the Indian Ocean, with the largest increase observed over the western Indian Ocean, followed by the north Bay of Bengal. These marine heatwaves occur as a result of background ocean warming in the Indian Ocean and also in response to El Niño events in the Pacific Ocean. Locally, a peak in solar radiation and a dip in evaporative cooling due to weak winds lead to the formation of these marine heatwaves. In the western Indian Ocean, the weak winds also reduce the heat transported by ocean currents from the near-equatorial regions toward the north, intensifying the marine heatwave. In turn, these marine heatwaves impact the monsoon by reducing the rainfall over the central Indian subcontinent while enhancing it over the southern peninsula.

## 1. Introduction

According to the Special Report on the Ocean and Cryosphere (SROCC) of the Intergovernmental Panel on Climate Change (IPCC), the global ocean will continue to warm during the 21st century (IPCC, 2019). By 2100, the ocean warming in the top 2,000 m is estimated to be 5–7 times higher under the business-as-usual scenario and 2–4 times higher under the low emission scenario, relative to the temperature reported since 1970 (*very likely*). Under this warming scenario, a rise in extreme temperature events in the ocean is also projected. Marine heatwaves (MHWs) are anomalous warm water events in response to the warming ocean—defined when the daily sea surface temperature (SST) exceeds the 90th percentile for 5 or more days (percentile threshold may vary and can be as high as the 99th percentile; Collins et al., 2019; Hobday et al., 2016; Schaeffer et al., 2017). MHWs are reported all over the world—the 2003 Mediterranean Sea MHW, 2011 West Australian MHW, 2013/15 Northeast Pacific MHW, 2015/16 Tasman Sea MHW are some of them. Numerous studies indicate that these anomalous

temperature events can cause habitat destruction due to coral bleaching, seagrass destruction, and loss of kelp forests, affecting the fisheries sector adversely (Collins et al., 2019; Mills et al., 2013).

Numerous studies have focused on the mechanisms leading to the genesis of MHWs. Most of the MHWs in Indo-Pacific regions are associated with climate modes like the El Niño, the Indian Ocean Dipole (IOD), and the Pacific Decadal Oscillation (PDO). The 2014–2015 MHW in the Northeast Pacific was associated with the PDO and the North Pacific Gyre Oscillation (NPGO), though local and remote ocean-atmospheric processes enhanced the conditions that led to an increase in SST in the Northeast Pacific (Bond et al., 2015; Capotondi et al., 2019; Joh & Lorenzo, 2017). Holbrook et al. (2019) suggested that the SST variability in the tropical Indian and Pacific oceans is related to the El Niño, IOD, and Indian Ocean warming.

Although several studies have explored the extreme temperature events in the Pacific and Atlantic oceans, there is hardly any in-depth research on MHWs in the Indian Ocean. In the southeast Indian Ocean, coral bleaching events due to the MHWs were reported, mostly associated with the El Niño and Madden-Julian Oscillations (MJO), riding over a global warming signal (Zhang et al., 2017). Studies have reported high SST conditions in the western Indian Ocean (Seychelles region), Arabian Sea, and the Bay of Bengal region (Andaman Sea) and also point out the coral bleaching events due to intense warming in these regions (Edward et al., 2018; Krishnan et al., 2011; Raj et al., 2018; Saji et al., 1999), suggesting that MHWs are behind these events.

The north Indian Ocean is a critical region because it supports large marine primary productivity, particularly during the South Asian summer monsoon (Roxy et al., 2016), and because of its role in modulating the intraseasonal and interannual variability of the monsoon (Roxy and Tanimoto, 2007; Roxy et al., 2013; Singh & Dasgupta, 2017). Understanding the genesis and evolution of MHWs during the summer monsoon season and their changes due to ocean warming is hence crucial. In this study, for the first time, we explore the genesis and trend of MHWs in the Indian Ocean during the Indian summer monsoon season (June–September). While many studies have addressed the impacts of MHWs on the marine ecosystem, there are hardly any studies investigating the impacts of MHWs on atmospheric circulation and rainfall. Hence, we investigate the interaction between MHWs and Indian summer monsoon circulation and rainfall.

#### 2. Data and Methods

#### **2.1. Data**

To identify the MHWs, we used the daily Optimum Interpolated Sea Surface Temperature (OISST) data set at a resolution of  $0.25^{\circ} \times 0.25^{\circ}$ , for the period 1982–2018 (Reynolds et al., 2007). In order to study the impact of MHWs on atmospheric conditions, we used the daily NOAA Interpolated Outgoing Longwave Radiation (OLR) measured from Advanced Very High-Resolution Radiometer (AVHRR) onboard NOAA polar-orbiting spacecraft (Liebmann & Smith, 1996), at a  $2.5^{\circ} \times 2.5^{\circ}$  resolution, for the period 1982–2018. The daily wind, latent heat flux, sensible heat flux, upward longwave radiation, downward solar radiation, surface net longwave radiation, and surface net solar radiation are obtained from the National Center for Environmental Prediction/ National Centers for Atmospheric Research (NCEP/NCAR) reanalysis data at a resolution of  $2.5^{\circ} \times 2.5^{\circ}$  (Kalnay et al., 1996). For the analysis of the rainfall variability over India, we used the Indian Meteorological Department (IMD) daily rainfall at a resolution of  $0.25^{\circ} \times 0.25^{\circ}$  (Pai et al., 2014). All the analyses are compared with European Centre for Medium-Range Weather Forecasts (ECMWF) ERA5 reanalysis product (Hersbach & Dee, 2016). In the main text, we use NCEP/NCAR products, and in the Supporting Information, we compare our results with ERA5 reanalysis products.

To investigate the ocean conditions during the MHWs, we obtained the ocean current data at  $1/3^{\circ}$  grid with a 5-day temporal resolution from the Ocean Surface Current Analysis Real-time (OSCAR; Dohan & Maximenko, 2010). The monthly potential temperature and salinity are obtained from ORAS4 at a horizontal resolution of  $1^{\circ} \times 1^{\circ}$  (Balmaseda et al., 2013). Apart from this, daily water temperature and salinity are obtained from the Hybrid Coordinate Ocean Model (HYCOM) reanalysis. We supplemented the heat budget analysis using the HYCOM reanalysis (Metzger et al., 2014) at a resolution of  $1/12^{\circ}$  for 1994–2015.



#### 2.2. Methods

#### 2.2.1. Detection of Marine Heatwaves

In this study, we identify MHWs from the OISST daily data sets, where the SST is above the 90th percentile (threshold) and persist for at least 5 days. Here, the 90th percentile and the baseline climatology of SST are calculated for each calendar day from the daily SSTs using an 11-day window centered on that particular day of the year, across all years, within the climatology period (1982-2018)-and are smoothed using a 31-day moving average (Gupta et al., 2020; Hobday et al., 2016; Oliver et al., 2018). The reason for considering an 11-day window is to create enough sample size (407 samples vs. 37 samples) for estimating the climatology and the 90th percentile at a daily timescale. This is important since due to day-to-day SST fluctuations, chances of a few outliers skewing the daily climatology or 90th percentile values are high. The 31-day moving average is used to obtain a smoothed climatology that is seasonally varying and does not contain any higher-frequency variability than the seasonal timescale (below 30 days). This seasonally varying 90th percentile will help us to identify the MHWs at any time of the year, regardless of the summer months (Oliver et al., 2018). We also utilize specific metrics to study and analyze the MHWs, such as the maximum intensity (maximum temperature anomaly during a particular MHW event), mean intensity (mean temperature anomaly during a specific MHW event), cumulative intensity (sum of temperature anomalies during a particular MHW event), duration (number of days between the starting and ending dates), and spatial extent (area covered by MHW event; Hobday et al., 2016). To study the spatial extent of the MHWs, we counted the grids where the MHW events occurred and multiplied them by the grid area. To estimate the trend in MHWs, we used the OISST data from 1982 to 2018 and calculated MHW at each grid, and then fit a linear trend model. Along with the annual values, we also prepare the MHW metrics for June to September.

Further, we did a composite analysis of wind, SST, latent heat flux, sensible heat flux, upward longwave radiation, and downward solar radiation during 10 days before and after the starting date of MHW events. This provides us with a comprehensive picture of the genesis of MHWs and describes the physical and dynamical processes involved with the MHW genesis over the particular regions. To study the interaction between the MHWs and summer monsoon circulation and rainfall, we have included a composite analysis of wind at 850 hPa, OLR, omega at 500 hPa, vertically integrated moisture flux, and rainfall over India during the MHWs.

#### 2.2.2. Mixed Layer Heat Budget

To study the local factors influencing the genesis of MHWs, we estimated the mixed layer heat budget using the temperature tendency equation. The temperature tendency equation gives us a comprehensive picture of factors that lead to the change in the ocean surface temperature (Rodrigues et al., 2019). The temperature tendency equation is given by

$$\partial T_m / \partial t = -v.\nabla T_m + Q_0 / \rho C_p H_m + Res$$
<sup>(1)</sup>

where  $\rho$  is the density of seawater (1,026 kg m<sup>-3</sup>), and  $C_p$  is the specific heat of seawater (3,902 J kg<sup>-1</sup> K<sup>-1</sup>), and  $T_m$  is the mixed layer temperature. Here we have used the SST as a mixed layer temperature because the vertically averaged mixed layer temperature is close to the SST values (Foltz et al., 2003).

 $\partial T_m/\partial t$  represents the rate of change of mixed layer temperature. The term -v.VT indicates the advection and v indicate the horizontal velocity vector (v vertically averaged at 15 m depth from OSCAR Ocean currents, representing the mixed layer).  $H_m$  means the mixed layer depth (MLD) in meters. Using the monthly potential temperature and salinity data of ORAS4, we calculated the monthly MLD as the depth at which the density changes by a threshold of 0.05 kg m<sup>-3</sup> for depth of 5 m. The monthly MLD is interpolated to daily to estimate the heat flux terms (Rodrigues et al., 2019).

$$Q_0 = Q_s + Q_b + Q_e + Q_h \tag{2}$$

 $Q_0$  is the net surface heat flux (W m<sup>-2</sup>), which includes the latent heat flux  $Q_e$ , and the sensible heat flux  $Q_h$  released from the ocean mixed layer, the net solar radiation received  $Q_s$ , and the net longwave radiation released  $Q_b$ . Finally, the *Res* include the remaining unresolved process, such as vertical diffusion, and entrainment. The heat budget analysis is also verified with HYCOM ocean current data sets. All the heat budget terms are calculated from 1994 to 2015 and averaged for the two study regions. Anomalies are derived from the 1994–2015 daily



climatology. Then we estimate the heat budget terms for the 5 days before the start date of MHWs in the selected regions.

#### 3. Results

To investigate the changes in MHWs in the Indian Ocean, we first select those regions in the tropical Indian Ocean that experienced an increasing trend in MHWs during 1982–2018. The result section is organized into five sections. Section 3.1 describes the MHWs in the Indian Ocean and their trends. In Section 3.2, we describe the seasonal climatology of MHWs for the regions where the trends are the largest. Section 3.3 addresses the role of dominant modes of climate variability on the MHWs in the tropical Indian Ocean. In Section 3.4, we focus on the genesis and evolution of MHWs during the southwest monsoon season (June–September) and then describes the interaction between MHWs and the monsoon rainfall in Section 3.5.

#### 3.1. Observed Trends in Marine Heat Waves in the Indian Ocean

The climatological mean distribution of MHW frequency during 1982–2018 is shown in Figures 1a and 1b, indicating that MHWs are generally observed in the northern parts of north Indian Ocean. The trend in the annual MHW frequency from 1982 to 2018 over the Indian Ocean is shown in Figure 1c. The western Indian Ocean region (41°E–56°E, 8°S–8°N) shows the most prominent trend in MHWs, annually, though climatologically it was not a region with frequent MHWs. The western Indian Ocean region has an increasing trend of 1.2–1.5 MHW events per decade (p < 0.01) at each grid point. The northern Arabian Sea also shows a similar increasing trend, though not well-defined. Meanwhile, in the eastern Indian Ocean and central equatorial Indian Ocean (80°E–110°E, 0°–20°S), the trend is about 0.4 MHW events per decade.

The boreal summer monsoon months (June–September) are immensely crucial for the Indian Ocean rim countries because the rainfall during this season supports the food, water, and energy security of this region (Beal et al., 2020; Wang et al., 2020). SST variability in the tropical Indian Ocean regulates the monsoon circulation and rainfall. Recognizing the importance of the SSTs during the northern summer monsoon months, we investigate the changes in MHWs during this season (June–September), for 1982–2018 (Figure 1d). Climatologically, during this season, MHWs generally occur in the northern parts of the north Indian Ocean (Figure 1b). During June–September, there are two regions where MHWs activities are rapidly increasing, in terms of the frequency and the area covered. One is in the western Indian Ocean region, where the annual trend in MHWs frequency is also maximum. The second one is in the northern Bay of Bengal (85°E–93°E, 15°N–23°N) region, where a well-defined trend in MHW frequency is observed, at its maximum during the summer monsoon season (Figure 1d). These two regions are experiencing an increasing trend of 0.5 events per decade during June–September. Apart from these two locations, the coastal northeastern Arabian Sea also shows a similar increasing trend. Chatterjee et al. (2022) links these coastal Arabian Sea MHWs to the increase in the mean SST in the recent decades and the Indian Ocean basin-wide warming in response to El-Niño events during their decaying phase.

We further examine the month-wise trend of the frequency of MHWs over western Indian Ocean and northern Bay of Bengal region. For the western Indian Ocean region (in Figure 1e), we find that the increasing trend of MHW frequency is present in all months. The trend is highest during October to December, with the maximum in November. The trend in MHW frequency is moderate during February to May and August to September. For the north Bay of Bengal region, the MHWs frequency trends are small in January to April and November, while large in May to September, and moderate in October and December (Figure 1f). Notably, the western Indian Ocean region is crucial because it is experiencing a significant increasing trend in the frequency of MHWs year-round, while the north Bay of Bengal region is experiencing a significant trend during summer monsoon months.

Henceforth, we focus our analysis on the western Indian Ocean (41°E–56°E, 8°S–8°N) and north Bay of Bengal (85°E–93°E, 15°N–23°N) regions, where the MHW trends are the largest. We identified individual MHW events based on the spatial mean SST time series over western Indian Ocean and north Bay of Bengal boxes in Figure 1d. There are 66 MHW events that occurred in the western Indian Ocean region during the 1982–2018 period, in which 21 events occurred during June–September. The north Bay of Bengal region witnessed 94 MHW events, and 34 out of these occurred during June–September. The time series of the annual and monsoon MHW frequency for western Indian Ocean and north Bay of Bengal are shown in Figures 1g and 1h. There is a significant rise in annual MHW frequency during 1982–2018 at a rate of 0.14 events per year for the western Indian Ocean and



# Journal of Geophysical Research: Oceans



**Figure 1.** Climatology and trend in MHW frequency (count/year) in the Indian Ocean during 1982–2018 (a), (c) annually and for (b), (d) June–September, using NOAA OISST data. Month-wise trend of MHW frequency in the (e) western Indian Ocean (WIO,  $41^{\circ}E-56^{\circ}E$ ,  $8^{\circ}S-8^{\circ}N$ ) and (f) north Bay of Bengal ( $85^{\circ}E-93^{\circ}E$ ,  $15^{\circ}N-23^{\circ}N$ ). Time series of the number of MHW days annually (blue bars), and during June–September (red bars), from 1982 to 2018 in (g) the western Indian Ocean (h) the north Bay of Bengal region. Trend lines in the figure are statistically significant (p < 0.05).





**Figure 2.** Month-wise climatology of MHWs. Month-wise distribution of (a) cumulative intensity (°C days), (b) mean intensity (°C), (c) duration (days), (d) maximum intensity (°C), (e) area (km<sup>2</sup>) and (f) the number of MHW events (month-wise sample size of MHWs, based on which the month-wise climatology is estimated), during 1982–2018 for the western Indian Ocean (WIO, 41°E–56°E, 8°S–8°N) and the north Bay of Bengal (BoB, 85°E–93°E, 15°S–23°N). The error bar represents the standard deviation.

0.09 events per year for the north Bay of Bengal. On the other hand, the MHW frequency during the monsoon season is increasing at a rate of 0.04 events per year in the western Indian Ocean region and 0.05 events per year in the north Bay of Bengal (Figure 1g and 1h). It is evident from the time series that the trend of MHW frequency in the north Bay of Bengal during June–September shows higher values compared to the western Indian Ocean region during the same period.

#### 3.2. Seasonal Climatology of Marine Heatwaves in the Tropical Indian Ocean

For the western Indian Ocean region, all the MHW metrics (except for the MHW events) show a double peak in the seasonal climatology signal during March and September months (Figure 2). During these 2 months, the duration of MHWs may reach 16 and 20 days, respectively. The mean intensity shows high seasonal variability. The mean intensity values during March and September are highest with mean values of 1.03°C and 1.1°C, respectively. In the case of maximum intensity, it is 1.3°C during the peak months. In the case of area covered by

the MHW events, it is 1.4 million km<sup>2</sup> during the peak months of March and September. Meanwhile, the number of MHW events show peaks during the May, August, and October months.

MHWs in the north Bay of Bengal are short-lived, and the cumulative intensity and area covered are smaller compared to the western Indian Ocean. The duration and cumulative intensity of MHWS in the north Bay of Bengal do not exhibit large seasonal variations. The cumulative intensity, duration, and area exhibit a peak during April, while the peak in the mean intensity, and maximum intensity is in May. Further, the number of events is maximum during May (14), June (14), and October (15) months. The cumulative intensity and duration in the peak month of April are 10°C days and 10 days, respectively. The mean intensity during the May month is 1.1°C, while the maximum intensity is 1.3°C. During April, the area is 0.4 million km<sup>2</sup>.

#### 3.3. Role of Indian Ocean Warming and Dominant Modes of Climate Variability on Marine Heatwaves

Recent research on MHWs (Frölicher et al., 2018; Holbrook et al., 2019; Laufkötter, et al., 2020) and the IPCC SROCC point out that MHW events are a manifestation of the global warming trend (Collins et al., 2019). A pattern correlation between the tropical SST trends and tropical MHW trends shows a significantly high correlation (r = 0.8 for annual values and r = 0.6 for boreal summer season) supporting this argument. Besides the global SST rise, various modes of climate variability favor MHW events by generating the necessary background conditions (Holbrook et al., 2019).

This section investigates the role of different climate modes on the occurrences of MHW events, particularly the association with El Niño, Indian Ocean basin-wide warming mode (IOB, Indo-Pacific warming generally occurring during the post- El Niño periods), Indian Ocean Dipole (IOD), and the North Atlantic Oscillation (NAO). Figure 3 shows the number of MHW days coincident with these climate modes or overlapping modes. An active phase of a climate mode is defined when the corresponding climate indices have a value larger than its long-term standard deviation. Note that while classifying the association between MHWs and the climate modes, MHWs during co-occurring climate modes are not counted for individual modes. For example, MHW days that occur when positive IOD and negative NAO co-occur are not considered under either the individual positive IOD-only or negative NAO-only groups. The absence of any climate modes during the MHW is indicated as "Nil". It is important to note that the metric "MHW days" is used instead of "MHW events" for investigating the association between the MHWs and different climate modes. This is because a greater number of MHW events in a season or year does not necessarily mean a correspondingly greater number of MHW days, as there may be a greater number of short MHW events in some years with short duration, and vice-versa.

Correlation and composite analysis of the yearly count of MHW days in the western Indian Ocean region with the annual mean SSTs show that these events are associated with the El Niño, IOB (Indian Ocean warming), and negative North Atlantic Oscillation (NAO; Figures 3a and 3b). Out of all the western Indian Ocean MHW days, the most significant number of MHW days occur under the Indian Ocean warming condition (Figures 3a and 3b). There are a significant number of MHW days when no climate modes are present (Figure 3b, gray bar). Regardless, there are a large number of MHW days under the positive IOD and negative NAO conditions. The spatial correlation between the MHW days in the western Indian Ocean and El Niño/IOD-like conditions (Figure 3c). The climate mode analysis shows that these MHW days occur either independent of the modes (Figure 3d, gray bar), or are associated with concurrent occurrences of positive IOD and negative NAO (Figure 3d, pink bar), positive IOD and El Niño (Figure 3d, dark-red bar) or during positive IOD only conditions, in that order. To summarize, Indian Ocean warming, El Niño, positive IOD, and negative NAO largely regulate the occurrence of MHWs in the western Indian Ocean.

Similar to the western Indian Ocean region, correlation and composite analysis of the yearly count of MHWs days in the north Bay of Bengal with the annual mean SST shows an association with the Indian Ocean warming, El Niño, and negative NAO (Figures 3e and 3f). Out of all the north Bay of Bengal MHW days, significant MHW days occur under the Indian Ocean warming condition. There are several MHW days associated with the negative NAO conditions following the IOB condition. Meanwhile, for June to September, the MHW days in the north Bay of Bengal appear to occur with warming in the north Indian Ocean (Figure 3g) and negative NAO pattern (Figure 3h).





**Figure 3.** Correlation between the total number of MHWs and global SST for (a), (e) annual and (c), (g) June to September in the western Indian Ocean (WIO) and the north Bay of Bengal (BoB). The bar-charts (b, d, f, h) indicate the total number of MHW days coinciding with climate modes. "Nil" means the MHW days that do not coincide with any of the climate modes. MHW days during co-occurring climate modes are not counted for individual modes.

In summary, occurrences of western Indian Ocean and north Bay of Bengal MHWs are more frequent while Indian Ocean warming is prominent basin wide (IOB) or when the western (for western Indian Ocean MHWs in summer) or northern (for Bay of Bengal MHWs in summer) exhibits warming. The western Indian Ocean SSTs also respond to the El Niño due to the subsidence of Walker circulation over these regions (Roxy et al., 2014; Xie et al., 2016). This subsidence inhibits deep convection and thereby suppresses the release of heat from the ocean to the atmosphere. Thus, the Indian Ocean attains its maximum temperatures under the IOB mode. That is, the background warming in the Indian Ocean has a strong influence on the MHWs in the Indian Ocean basin on an annual timescale. During a positive IOD event, the western Indian Ocean becomes warmer than its climatology (Murtugudde & Busalacchi, 1999; Saji et al., 1999). June–September MHWs in the western Indian Ocean region are often associated with these positive IODs (Holbrook et al., 2019). The negative NAO phases are also associated with MHWs over the north Bay of Bengal region. Holbrook et al. (2019) and Mishra et al. (2012) show an association between negative NAO and north Indian Ocean warming. However, the mechanism between the negative NAO and north Indian Ocean SSTs needed to be explored in detail. Though the occurrences of MHWs in the







**Figure 4.** (a) The lead-lag correlation between SST and wind speed (red line), latent heat flux (orange line), sensible heat flux (green line), upward long wave radiation (black line), OLR (blue line) in the western Indian Ocean, estimated from the 30 days before and 30 days after the genesis of MHW events. The dotted lines represent the correlation significant at 95% confidence level based on a student's *t*-test. Composite evolution of (b) SST (°C), latent heat (LH, W m<sup>-2</sup>), wind (m s<sup>-1</sup>), OLR (W m<sup>-2</sup>), shortwave radiation (SW, W m<sup>-2</sup>), sensible heat (SH, W m<sup>-2</sup>), and longwave (LW, W m<sup>-2</sup>), in the western Indian Ocean before and during MHWs for the period 1982–2018, using NCEP/ NCAR reanalysis datasets. Upward arrow represents the exchange of flux from ocean to atmosphere and downward arrow represents the exchange of flux from atmosphere to ocean.

Indian Ocean exhibit potential association with the climate modes described here, we do not investigate the mechanisms through which their causal relationships are manifested.

# **3.4.** Local Ocean-Atmospheric Interaction Leading to the Genesis and Evolution of MHW Events

In this section we explore the role of local ocean-atmospheric conditions leading to the initiation and the persistence of MHWs in the Indian Ocean. It is observed that for the western Indian Ocean MHWs, the downward solar radiation leads the SST peak by ~10 days. However, this lead-lag relation is weak (r = 0.13). The increased downward solar radiation due to clear sky conditions may increase SST in the western Indian Ocean region but is not the primary reason for MHW genesis. Meanwhile, the reduced wind speed and evaporation (denoted by the negative wind anomaly and reduced upward latent heat flux anomaly in Figure 4b) leads the SST anomalies by  $\sim$ 3 days (r = -0.24, p < 0.05), assisting the MHW formation. Besides the air-sea fluxes, reduced wind speed also impacts the ocean current over the western Indian Ocean Region. The ocean currents in the western Indian Ocean region (driven by strong cross-equatorial monsoon winds) are usually strong and transport heat away from the equator. Hence, a weakening of the cross-equatorial winds can potentially reduce ocean heat transport by accumulating the heat in this region, enhancing the MHWs.

To comprehend the spatial distribution of the lead-lag relationship, composites of different variables from 10 days before and after the MHW genesis date are presented in Figure 4b. The SST anomalies show a dipole pattern with cold anomalies at 15°N and warm anomalies at the equator prior to the genesis of western Indian Ocean MHWs. In Figure 4a, the region of cold SST anomalies can be observed north of the western Indian Ocean region 10 days prior to the MHWs initiation. The warm SST anomalies near the equator gradually become prominent as the MHW initiates. Interestingly, the cold SST anomalies (up to  $-0.5^{\circ}$ C) at the north of the western Indian Ocean region is maintained consistently from about a month (30 days) to almost a week (6 days) prior to the MHWs genesis. The meridional SST difference between the north (15°N) and western Indian Ocean region is also maintained during the same period prior to the genesis of MHWs (Figure S4c, S4d in Supporting Information S1). These cold SST anomalies at 15°N and the meridional SST difference may have a significant role in reducing the strength of cross-equatorial wind flow (Figure 4b and Figure S4 in Supporting Information S1). The reduced wind speed (up to -1 m/s during the pentad leading to the MHW onset) not only reduces the air-sea fluxes (reduce evaporation; Figure 4b and Figure S4 in Supporting Information S1) but also affects the strength of ocean currents, leading to decreased heat transport from the equator toward the north. These factors contribute to the MHW formation in the particular region. Ten days prior to the MHWs genesis, the clear sky condi-

tion over the western Indian Ocean region due to the weak cross-equatorial flow and less evaporation increases downward solar radiation. The increase in downward solar radiation further increases the SST. Therefore, from the lead-lag time series and the spatial composites, we identify the role of reduced heat transport, decreased evaporation from the ocean surface, and increased downward short-wave radiation behind the MHWs genesis over the western Indian Ocean region. We find the release of upward sensible heat flux and latent heat flux from the ocean after 3–5 days of the MHWs genesis reduce the ocean temperature and thereby the MHW intensity.

To explore the ocean dynamics involved in the genesis of MHWs over the western Indian Ocean region, we performed an oceanic mixed layer heat budget analysis (Figure 6a, 6b and 6c). We calculated the composites of the





**Figure 5.** (a) The lead-lag correlation between SST and wind speed (red line), latent heat flux (orange line), sensible heat flux (green line), upward long wave radiation (black line), OLR (blue line) in the north Bay of Bengal, estimated from the 30 days before and 30 days after the genesis of MHW events. The dotted lines represent the correlation significant at 95% confidence level based on a student's *t*-test. Composite evolution of (b) SST (°C), latent heat (LH, W m<sup>-2</sup>), wind (m s<sup>-1</sup>), OLR (W m<sup>-2</sup>), shortwave radiation (SW, W m<sup>-2</sup>), sensible heat (SH, W m<sup>-2</sup>), and longwave (LW, W m<sup>-2</sup>), in the western Indian Ocean before and during MHWs for the period 1982–2018, using NCEP/NCAR reanalysis datasets. Upward arrow represents the exchange of flux from atmosphere to ocean.

SH (W m-2)

OLB (W m<sup>-2</sup>)

LH (W m<sup>-2</sup>)

heat budget terms for the 5 days before the onset of MHWs (5-day average). The heat budget analysis is carried out for the MHWs between 1994 and 2015, based on the availability of ocean potential temperature and salinity data. Figures 6a and 6b shows the role of the advection terms and the heat flux terms on the rate of change of SST in the western Indian Ocean region. In the advection term, the heat transport by the zonal ocean currents (the zonal advection term) is dominant compared to the meridional component.

In comparison to the advection term, the heat flux term is found to be contributing more to the temperature change. In the net heat flux term, the net solar radiation received in the mixed layer and the latent heat stored (due to less evaporation) in the mixed layer are the major contributors. The solar heating of the ocean surface and decreased evaporative cooling favors the genesis of MHWs in the western Indian Ocean region. The time-integrated heat budget terms in Figure 6c reveal the time evolution of different heat budget parameters.

In the north Bay of Bengal region, reduced wind speed anomalies and evaporation lead the MHWs genesis by  $\sim 2$  days (r = -0.38; Figure 5a). Due to weakened wind speed, evaporative cooling over the ocean surface becomes less, and hence the SST increases. An increased OLR and downward solar radiation lead SST by ~4 days with r values 0.28 and 0.31, respectively. Hence, the reduced evaporative cooling and increased incoming solar insolation (clear sky conditions) jointly raise the SSTs over the north Bay of Bengal region leading to the MHWs. From the spatial composites in Figure 5b, the SST shows a rapid increase coinciding with the MHW genesis day. It is seen that the northeasterly winds establish in the north Bay of Bengal region 5 days before the onset of MHW and weaken the southwesterlies over the particular region. This reduced wind speed leads to a decrease in evaporation at the ocean surface, in turn reducing the release of upward latent heat flux from the ocean. A negative latent heat flux anomaly in the north Bay of Bengal region is observed 5 days before the onset of MHWs, confirming these results. Moreover, there is an increased downward solar radiation and increased OLR 5 days before the genesis of MHWs, peaking on the starting day of MHWs. We can infer that the decreased upward latent heat flux due to the decreased wind speed and the clear sky condition (increased downward short-wave radiation) contribute to the MHW genesis in the north Bay of Bengal region. After 3 days of the MHW genesis, there is a release of sensible heat and upward longwave radiation from the ocean, reducing the ocean temperature and helping terminate MHWs events.

It is worth noting that (Figure 6d, 6e and 6f) in the northern Bay of Bengal, the temperature change is mainly attributed to the net heat flux, and there is less contribution from the heat advection term. The latent heat (due to decreased evaporation) and net solar radiation have a prominent role among the heat flux terms. Meanwhile, the role of advection is minimal here. In a nutshell, it can be inferred that the weak wind causes decreased evaporative cooling of the ocean, while the increased solar heating is also accompanying the temperature rise in this region. Figure 6f represents the time-integrated heat budget terms. It is evident that for the north Bay of Bengal region, the MHW duration is comparatively less than the western Indian Ocean region.

In summary, the role of reduced wind speed and evaporation is more prominent in north Bay of Bengal MHWs, than the western Indian Ocean MHWs.

The lead-lag and heat budget analysis here using NCEP reanalysis are cross-validated using ERA5 reanalysis products in the Supporting Information (Figure S2 & S3 in Supporting Information S1). In the lead-lag analysis

SST (°C)



# Journal of Geophysical Research: Oceans





the ERA5 and NCEP show similar results. The significant lead-lag correlations of both the data set are similar, which supports the proposed lead-lag relation of SST and other atmospheric conditions, except the correlation with OLR. In case of heat budget analysis, the ERA5 (Figure S5 in Supporting Information S1) and the NCEP (Figure 6) show a similar conclusion, indicating that the downward solar radiation and latent heat fluxes contribute largely to the peak in SST. However, there are some disagreements also. For example, in the western Indian Ocean region, the ERA5 indicates that the temperature advection by ocean currents is a larger contributor than the net heat flux (Figure S5a and S5b in Supporting Information S1). Also, for the both regions, the contribution of downward solar radiation to the net heat fluxes are large in the ERA5 (Figure S5 in Supporting Information S1) as compared to the NCEP (Figure 6) reanalysis. The differences in the heat flux terms contribution may be due to the nature of the reanalysis products.

#### 3.5. Interaction of Marine Heatwaves With the Indian Summer Monsoon

The analysis in this section is focused on the summer monsoon rainfall variations during the MHWs over both the western Indian Ocean and north Bay of Bengal regions. We analyzed the mean pattern of SST, vertical velocity (omega at 500 hPa), wind (at 850 hPa), OLR, and vertically integrated moisture flux and the rainfall. It is fascinating that the MHWs over the two regions have the opposite impacts on Indian summer monsoon rainfall.

The SST anomaly composite during western Indian Ocean MHWs shows a warming pattern in this region. The SST anomalies vary from 0.9°C to 1.2°C (Figure 7a). Moreover, negative omega (vertical velocity) anomalies at 500 hPa and OLR confirm enhanced convective activity during MHW days over the western Indian Ocean MHW region. Higher SST triggers the ascending motion of air and convection (Figure 7c, Figure S6a in Supporting Information S1). At the same time, the circulation pattern shows strong northeasterly wind anomalies, indicating that the mean southwesterly monsoon winds are weak during this time (Figure S6a in Supporting Information S1). The anomalous Hadley circulation during western Indian Ocean MHW events shows increased convective activities over the equatorial Indian Ocean, while the descending branch occurs over the Indian subcontinent (Figure 7g). This indicates that the western Indian Ocean MHWs are conducive for inducing dry conditions over the Indian landmass. The vertically integrated moisture flux divergence analysis contributes a more precise picture, suggesting an anomalous moisture flux divergence over the Indian subcontinent, with the moisture transported away toward the western Indian Ocean region (Figure 7e). The mean pattern of rainfall over Indian landmass agrees with the results, with reduced rainfall across most of the subcontinent where the monsoon westerlies are prominent (Figure 7i).

In Figure 7b, the higher SST anomalies in the north Bay of Bengal region are apparent during June–September MHWs. The positive omega anomalies at 500 hPa and the positive OLR values imply dry subsidence and absence of cloud cover (clear sky condition) in this region during strong MHW events (Figure 7d, Figure S6b in Supporting Information S1). However, negative OLR anomalies (above  $-15W m^{-2}$ ) and negative omega anomalies at 500 hPa level over the western India and adjacent ocean region suggest enhanced convection and strong ascending motion in these regions. The vertically integrated moisture flux anomalies (Figure 7f) indicate a convergence of the wind in the same region where the OLR and omega anomalies are negative.

Figure 7h shows the mean meridional circulation averaged over the  $41^{\circ}E-100^{\circ}E$  for the north Bay of Bengal region during the June–September MHW days for the period 1982–2018. Over the southern part of the Indian subcontinent (0–15°N), there is increased convection during MHW days, supporting the previous results that show an increased convection in the vertical velocity over the region, while there is a suppressed convection at the 20°N due to the dry subsidence over north Bay of Bengal. We find an anomalous moisture flux convergence over the Indian landmass and anomalous moisture flux divergence over the north Bay of Bengal, which provides the cause for the observed rainfall pattern. The moisture transport from the warm north Bay of Bengal region by the anomalous easterly winds plays a crucial role in enhancing the rainfall over the south peninsular Indian subcontinent. There is a convergence of southwesterly monsoon winds and anomalous easterly winds from the Bay of Bengal at 850 hPa over the south-west Indian subcontinent (Figure S6b in Supporting Information S1). The moisture convergence introduces the convective instability over this region. To summarize, there is a significant relationship between the MHWs in the northern Bay of Bengal and increased summer monsoon rainfall over southwest India.

#### 4. Conclusion

Here we have conducted a detailed investigation of the trends and genesis of MHW events in the tropical Indian Ocean—and its interaction with the Indian summer monsoon (Figure 8). We find that the trend in annual frequency of MHWs is relatively higher in the western equatorial Indian Ocean ( $41^{\circ}E-56^{\circ}E$ ,  $8^{\circ}S-8^{\circ}N$ ), which experienced 66 events during the 1982–2018 periods. We further studied the trend in MHW frequency during the Indian summer monsoon period (June–September) and find that the trend in MHWs is prominent in the western Indian Ocean and the north Bay of Bengal ( $85^{\circ}E-93^{\circ}E$ ,  $15^{\circ}N-23^{\circ}N$ ). The north Bay of Bengal experienced 94 MHW events, wherein 34 events occurred from June to September, while in the western Indian Ocean region, it was 21 events.





**Figure 7.** Composite anomaly of daily (a), (b) sea surface temperature anomalies (SST, °C), (c), (d) vertical velocity anomalies (omega at 500 hPa, Pa s<sup>-1</sup>), (e), (f) moisture flux convergence (kg m<sup>-1</sup> s<sup>-1</sup>), (g), (h) meridional circulation over  $41^{\circ}E-100^{\circ}E$  (vertical velocity, Pa s<sup>-1</sup>) and (i), (j) rainfall (mm day<sup>-1</sup>) during MHW days in the western Indian Ocean and north Bay of Bengal, during June–September, for 1982–2018. Stippling (black dots) indicates anomaly values significant at 95% confidence level.





Marine heatwaves in the Indian Ocean and their impact on the monsoon

Figure 8. Schematic diagram depicting the factors leading to the genesis of MHW in (a) the western Indian Ocean and (b) the north Bay of Bengal, and its impact on the Indian summer monsoon rainfall. The legend alongside the thermometer indicates the association with ocean warming and various climate modes.

We further investigate the climate modes driving the MHWs over these two regions. It is observed that the longterm persistence of MHWs in the western Indian Ocean region is associated with the background SST warming primarily identified with the Indian Ocean basin-wide warming (IOB) and also the El Niño, IOD, and NAO. Roxy et al. (2014) and Abish et al. (2018) show that the western Indian Ocean is warming at a rapid pace since the 1950s, largely in response to anthropogenic warming but manifested through an asymmetry in the ENSO impact—whereby El Niño events cause anomalous warming in the western Indian Ocean and La Niña events fail to cool the region. They also found that the ENSO events are more intense in the recent decades and this result in a positive SST skewness. Han et al. (2010) identifies a change in the local Hardly and Walker circulation that affect the distribution of sea level and temperature in the in the Indian Ocean, with an increase in the western Indian Ocean. Regardless, the role of mean SST rise in the western Indian Ocean is obvious (Roxy et al., 2020). Figure S7 in Supporting Information S1 shows an increase in the SST in the western Indian Ocean after year 2000 (both for annual and June-September) in comparison with pre-2000 period, indicating the role of background warming on the increase in MHWs in the western Indian Ocean. The background warming and the MHWs potentially decrease the summer monsoon rainfall over the central and north east India post-2000, as observed by the rainfall difference between the two periods (Figure S7c in Supporting Information S1).

For the north Bay of Bengal MHWs, the longer duration of these events is linked to the IOB, El Niño, and NAO. Earlier studies have pointed out that El Niño, IOD and IOB are major contributors to anomalous warm SST events in the tropical ocean (Holbrook et al., 2019). Notably, the SST trend due to global warming is manifested through the increased number of MHW events.

We investigated the seasonal climatology of MHW metrics in both the selected regions. For the western Indian Ocean region, it is found that all the MHWs metrics (except for the MHW events) have a double peak in the seasonal climatology during March and September months. The highest number of western Indian Ocean MHW events occur during May, August, and October. As compared to the MHWs in the western Indian Ocean, the MHWs in the north Bay of Bengal are short-lived, and the cumulative intensity and the area covered are smaller. The MHWs in the north Bay of Bengal do not exhibit large seasonal variation in the duration and cumulative intensity. The highest number of north Bay of Bengal MHWs events occur during May, June, and October months (14, 14, and 15 events, respectively). Interestingly, these months (May, early June, and October) are when the cyclones are active in the Bay of Bengal. Warm SSTs set an ideal precondition for cyclones, and it is hence necessary to closely monitor the basin to understand how these MHWs are interacting with the cyclones.

Focusing on the factors influencing the genesis of MHWs events in the tropical Indian Ocean, the results here show that in the western Indian Ocean region, MHWs are formed due to a relaxation of winds prior to the event.

This tranquility in the winds is making the SST rise through two ways—through a decrease in the evaporative cooling at the ocean surface, and through a weakened transport of the heat by ocean currents. Apart from this, increased solar heating of the ocean surface due to reduced cloud cover also helps the formation of a warm pool in the western Indian Ocean region. We find that solar insolation plays a major role in MHW genesis from ocean mixed layer heat budget analysis. The reduced upward latent heat flux and zonal heat advection are the other major contributors. Interestingly, we observe an occurrence of cold SST (north of the western Indian Ocean region) region 35–40 days prior to western Indian Ocean MHWs and a relatively low SST gradient. This low SST gradient and cold pool may be the reason for the weakening of cross-equatorial flow. The genesis of MHWs in the Tasman Sea and China's marginal seas also had a similar role in ocean dynamics (Oliver et al., 2017; Yao et al., 2020).

For the north Bay of Bengal region, it is observed that the reduction in wind speed and thereby reduction in the release of latent heat is one of the major causes for MHW genesis. Along with this, the increased solar radiation at the ocean surface also plays a key role. However, unlike the western Indian Ocean region, the signature of heat advection behind the genesis of MHWs is absent in the case of the north Bay of Bengal. In this study, we find that the western Indian Ocean MHWs have a longer persistence time, and the process of genesis is also slower than the MHWs in the north Bay of Bengal.

The MHWs in the western Indian Ocean and the north Bay of Bengal have strong links to the observed variability of the Indian summer monsoon rainfall. We find that both the MHWs are associated with the drying conditions over the central Indian subcontinent. However, there is a significant increase in the rainfall over south peninsular India in response to the MHWs in the north Bay of Bengal. Climate model projections suggest further warming of the Indian Ocean in the future, which will very likely intensify the MHWs (Collins et al., 2019) and their impact on the monsoon rainfall. Since the frequency, intensity, and area of MHWs are increasing, it is essential that we closely monitor and forecast these events in advance in order to mitigate their impacts.

### **Data Availability Statement**

The Optimum Interpolated Sea Surface Temperature (OISST) and the Outgoing Longwave Radiation (OLR) are available from National Oceanic and Atmospheric Administration (NOAA; Reynolds et al., 2007). The daily wind, latent heat flux, sensible heat flux, upward longwave radiation, downward solar radiation, surface net longwave radiation, and surface net solar radiation are available from the National Center for Environmental Prediction/National Centers for Atmospheric Research (NCEP/NCAR) reanalysis data (Kalnay et al., 1996). For the analysis of the rainfall variability over the India, we used daily rainfall available from the Indian Meteorological Department (Pai et al., 2014). The ocean current data is available from the Ocean Surface Current Analysis Real-time (OSCAR), provided by the Earth Space Research (Dohan & Maximenko, 2010).

#### Acknowledgments

J. S. S. acknowledges the support of Dr. P. O. Nameer, Dean of College of Climate Change and Environmental Sciences (CCES) and Dr. M. Rajeevan, Secretary to the Ministry of Earth Sciences (MoES), for supporting her visit to the Indian Institute of Tropical Meteorology (IITM).

#### References

Abish, B., Cherchi, A., & Ratna, S. B. (2018). ENSO and the recent warming of the Indian Ocean. International Journal of Climatology, 38(1), 203–214. https://doi.org/10.1002/joc.5170

Balmaseda, M. A., Trenberth, K. E., & Källén, E. (2013). Distinctive climate signals in reanalysis of global ocean heat content. *Geophysical Research Letters*, 40, 1754–1759. https://doi.org/10.1002/grl.50382

Beal, L. M., Vialard, J., Roxy, M. K., Li, J., Andres, M., Annamalai, H., et al. (2020). A Road map to IndOOS-2: Better observations of the rapidly warming Indian ocean. Bulletin of the American Meteorological Society, 101(11), E1891–E1913.

Bond, N. A., Cronin, M. F., Freeland, H., & Mantua, N. (2015). Causes and impacts of the 2014 warm anomaly in the NE Pacific. *Geophysical Research Letters*, 42(9), 3414–3420. https://doi.org/10.1002/2015gl063306

Capotondi, A., Sardeshmukh, P. D., Di Lorenzo, E., Subramanian, A. C., & Miller, A. J. (2019). Predictability of US west coast ocean temperatures is not solely due to ENSO. *Scientific Reports*, 9(1), 1–10. https://doi.org/10.1038/s41598-019-47400-4

Chatterjee, A., Anil, G., & Shenoy, L. R. (2022). Marine heatwaves in the Arabian Sea. Ocean Science Discussions, 1-27. In Review

Collins, M., Sutherland, M., Bouwer, L., Cheong, S.-M., Frölicher, T., Jacot Des Combes, H., et al. (2019). Extremes, abrupt changes and managing risk. In H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Eds.), *IPCC special report on the* ocean and cryosphere in a changing climate.

Dohan, K., & Maximenko, N. (2010). Monitoring ocean currents with satellite sensors. Oceanography, 23, 94–103. https://doi.org/10.5670/ oceanog.2010.08

Edward, J. P., Mathews, G., Raj, K. D., Laju, R. L., Bharath, M. S., Arasamuthu, A., et al. (2018). Coral mortality in the Gulf of Mannar, southeastern India, due to bleaching caused by elevated sea temperature in. *Current Science*, *114*(9), 1967. https://doi.org/10.18520/cs/v114/ i09/1967-1972

Foltz, G. R., Grodsky, S. A., Carton, J. A., & McPhaden, M. J. (2003). Seasonal mixed layer heat budget of the tropical Atlantic Ocean. *Journal of Geophysical Research Oceans*, 108(C5). https://doi.org/10.1029/2002jc001584

- Frölicher, T. L., Fischer, E. M., & Gruber, N. (2018). Marine heatwaves under global warming. Nature, 560, 360–364. https://doi.org/10.1038/ s41586-018-0383-9
- Gupta, A. S., Thomsen, M., & Benthuysen, J. A. (2020). Drivers and impacts of the most extreme marine heatwave events. Scientific Reports, 10, 19359. https://doi.org/10.1038/s41598-020-75445-3
- Han, W., Meehl, G. A., Rajagopalan, B., Fasullo, J. T., Hu, A., Lin, J., et al. (2010). Patterns of Indian Ocean sea-level change in a warming climate. *Nature Geoscience*, *3*(8), 546–550.
- Hersbach, H., & Dee, D. J. E. N. (2016). ERA5 reanalysis is in production. ECMWF Newsletter, 147(7), 5-6.
- Hobday, A. J., Alexander, L. V., Perkins, S. E., Smale, D. A., Straub, S. C., Oliver, E. C. J., et al. (2016). A hierarchical approach to defining marine heatwaves. *Progress in Oceanography*, 141, 227–238. https://doi.org/10.1016/j.pocean.2015.12.014
- Holbrook, N. J., Scannell, H. A., Gupta, A. S., Benthuysen, J. A., Feng, M., Oliver, E. C. J., et al. (2019). A global assessment of marine heatwaves and their drivers. *Nature Communications*, 10, 1–13. https://doi.org/10.1038/s41467-019-10206-z
- IPCC. (2019). In H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Eds.), IPCC special report on the ocean and cryosphere in a changing climate.
- Joh, Y., & Lorenzo, E. D. (2017). Increasing coupling between NPGO and PDO leads to prolonged marine heatwaves in the northeast Pacific. Geophysical Research Letters, 44(11), 663671–663777. https://doi.org/10.1002/2017GL075930
- Kalnay, E., Kanamitsu, M., Kistler, R., Collins, W., Deaven, D., Gandin, L., et al. (1996). The NCEP/NCAR 40-year reanalysis project. Bulleltin of American Meteorological Society, 77(3), 437–472. https://doi.org/10.1175/1520-0477(1996)077<0437:tnyrp>2.0.co;2
- Krishnan, P., Roy, S. D., George, G., Srivastava, R. C., Anand, A., Murugesan, S., et al. (2011). Elevated sea surface temperature during May 2010 induces mass bleaching of corals in the Andaman. *Current Science*, 100, 111–117.
- Laufkötter, C., Zscheischler, J., & Frölicher, T. L. (2020). High-impact marine heatwaves attributable to human-induced global warming. *Science*, 369(6511), 1621–1625.
- Liebmann, B., & Smith, C. A. (1996). Description of a complete (interpolated) outgoing longwave radiation dataset. Bulletin of the American Meteorological Society, 77(6), 1275–1277.
- Metzger, E. J., Smedstad, O. M., Thoppil, P. G., Hurlburt, H. E., Cummings, J. A., Wallcraft, A. J., et al. (2014). US Navy operational global ocean and Arctic ice prediction systems. *Oceanography*, 27(3), 32–43. https://doi.org/10.5670/oceanog.2014.66
- Mills, K. E., Pershing, A. J., Brown, C. J., Chen, Y., Chiang, F.-S., Holland, D. S., et al. (2013). Fisheries management in a changing climate: Lessons from the 2012 ocean heat wave in the Northwest Atlantic. *Oceanography*, 26, 191–195. https://doi.org/10.5670/oceanog.2013.27
- Mishra, V., Smoliak, B. V., Lettenmaier, D. P., & Wallace, J. M. (2012). A prominent pattern of year-to-year variability in Indian Summer Monsoon Rainfall. Proceedings of the National Academy of Sciences of the United States of America, 109(19), 7213–7217. https://doi.org/10.1073/ pnas.1119150109
- Murtugudde, R., & Busalacchi, A. J. (1999). Interannual variability of the dynamics and thermodynamics of the tropical Indian Ocean. Journal of Climate, 12(8), 2300–2326. https://doi.org/10.1175/1520-0442(1999)012<2300:ivotda>2.0.co;2
- Oliver, E. C., Benthuysen, J. A., Bindoff, N. L., Hobday, A. J., Holbrook, N. J., Mundy, C. N., et al. (2017). The unprecedented 2015/16 Tasman Sea marine heatwave. *Nature Communications*, 8(1), 1–12. https://doi.org/10.1038/ncomms16101
- Oliver, E. C., Donat, M. G., Burrows, M. T., Moore, P. J., Smale, D. A., Alexander, L. V., et al. (2018). Longer and more frequent marine heatwaves over the past century. *Nature Communications*, 9(1), 1–12. https://doi.org/10.1038/s41467-018-03732-9
- Pai, D. S., Sridhar, L., Rajeevan, M., Sreejith, O. P., Satbhai, N. S., & Mukhopadhyay, B. (2014). Development of a new high spatial resolution (0.25 × 0.25) long period (1901–2010) daily gridded rainfall data set over India and its comparison with existing data sets over the region. *Mausam*, 65(1), 1–18.
- Raj, K. D., Mathews, G., Bharath, M. S., Laju, R. L., Kumar, P. D., Arasamuthu, A., & Edward, J. P. (2018). Cushion star (Culcitaschmideliana) preys on coral polyps in Gulf of Mannar, southeast India. *Marine and Freshwater Behaviour and Physiology*, 51(2), 125–129. https://doi.org /10.1080/10236244.2018.1473013
- Reynolds, R. W., Smith, T. M., Liu, C., Chelton, D. B., Casey, K. S., & Schlax, M. G. (2007). Daily high-resolution-blended analyses for sea surface temperature. *Journal of Climate*, 20(22), 5473–5496. https://doi.org/10.1175/2007jcli1824.1
- Rodrigues, R. R., Taschetto, A. S., Gupta, A. S., & Foltz, G. R. (2019). Common cause for severe droughts in South America and marine heatwaves in the South Atlantic. *Nature Geoscience*, 12, 620–626. https://doi.org/10.1038/s41561-019-0393-8
- Roxy, M., & Tanimoto, Y. (2007). Role of SST over the Indian Ocean in influencing the intraseasonal variability of the Indian summer monsoon. Journal of the Meteorological Society of Japan. Series II, 85(3), 349–358. https://doi.org/10.2151/jmsj.85.349
- Roxy, M., Tanimoto, Y., Preethi, B., Terray, P., & Krishnan, R. (2013). Intraseasonal SST-precipitation relationship and its spatial variability over the tropical summer monsoon region. *Climate Dynamics*, 41(1), 45–61. https://doi.org/10.1007/s00382-012-1547-1
- Roxy, M. K., Gnanaseelan, C., Parekh, A., Chowdary, J. S., Singh, S., Modi, A., et al. (2020). Indian Ocean warming. In R. Krishnan, J. Sanjay, C. Gnanaseelan, M. Mujumdar, & A. Kulkarni (Eds.), Assessment of climate change over the Indian region. Springer. https://doi.org/10.1007/978-981-15-4327-2\_10
- Roxy, M. K., Modi, A., Murtugudde, R., Valsala, V., Panickal, S., Prasanna Kumar, S., et al. (2016). A reduction in marine primary productivity driven by rapid warming over the tropical Indian Ocean. *Geophysical Research Letters*, 43(2), 826–833. https://doi.org/10.1002/2015g1066979
- Roxy, M. K., Ritika, K., Terray, P., & Masson, S. (2014). The curious case of Indian ocean warming. Journal of Climate, 27, 8501–8509. https://doi.org/10.1175/JCLI-D-14-00471.1
- Saji, N. H., Goswami, B. N., Vinayachandran, P. N., & Yamagata, T. (1999). A dipole mode in the tropical Indian Ocean. *Nature*, 401(6751), 360–363. https://doi.org/10.1038/43854
- Schaeffer, A., Gramoulle, A., Roughan, M., & Mantovanelli, A. (2017). Characterizing frontal eddies along the East Australian Current from HF radar observations. *Journal of Geophysical Research Oceans*, 122, 3964–3980. https://doi.org/10.1002/2016JC012171
- Singh, C., & Dasgupta, P. (2017). Unraveling the spatio-temporal structure of the atmospheric and oceanic intra-seasonal oscillations during the contrasting monsoon seasons. Atmospheric Research, 192, 48–57. https://doi.org/10.1016/j.atmosres.2017.03.020
- Wang, B., Biasutti, M., Byrne, M. P., Castro, C., Chang, C. P., Cook, K., et al. (2020). Monsoons climate change assessment. *Bulletin of the American Meteorological Society*.
- Xie, S. P., Kosaka, Y., Du, Y., Hu, K., Chowdary, J. S., & Huang, G. (2016). Indo-western Pacific Ocean capacitor and coherent climate anomalies in post-ENSO summer: A review. *Advances in Atmospheric Sciences*, 33(4), 411–432. https://doi.org/10.1007/s00376-015-5192-6
- Yao, Y., Wang, J., Yin, J., & Zou, X. (2020). Marine heatwaves in China's marginal seas and adjacent offshore waters: Past, present, and future. Journal of Geophysical Research Oceans, 125, e2019JC015801. https://doi.org/10.1029/2019JC015801
- Zhang, N., Feng, M., Hendon, H. H., Hobday, A. J., & Zinke, J. (2017). Opposite polarities of ENSO drive distinct patterns of coral bleaching potentials in the southeast Indian Ocean. *Scientific Reports*, 7. https://doi.org/10.1038/s41598-017-02688-y